Author's response to reviews

Title: p16INK4a hypermethylation and p53, p16 and MDM2 protein expression in Esophageal Squamous Cell Carcinoma

Authors:

Noushin Taghavi (taghavi.noushin@gmail.com)
Firouzeh Biramijamal (f.birami@nigeb.ac.ir)
Masoud Sotoudeh (masoudso88@gmail.com)
Hooman Khademi (khademi@ams.ac.ir)
Reza Malekzadeh (malek@ams.ac.ir)
Omeed Moaven (omeedmoaven@gmail.com)
Bahram Memar (memarb@mums.ac.ir)
Azade A'radi (a_arabi57@yahoo.com)
Mohammad Reza Abbaszadegan (abbaszadeganmr@mums.ac.ir)

Version: 6 Date: 31 March 2010

Author's response to reviews: see over
Dear Editor,

As for the final proof of the manuscript # 2133304791325108, I am resubmitting the revised version which is the final draft containing the following change in the title page:

The order of the affiliations of the first author is changed and the authors detailed are changed to the following:

Noushin Taghavi¹,², Firouzeh Biramijamal³, Masoud Sotoudeh², Hooman Khademi², Reza Malekzadeh², Omeed Moaven³, Bahram Memar⁴, Azadeh A'rabi³, Mohammad Reza Abbaszadegan³

1. National Institute of Genetic Engineering and Biotechnology (NIGEB), Tehran, Iran
2. Digestive Disease Research Center (DDRC), Tehran University of Medical Sciences, Tehran, Iran
3. Division of Human Genetics, Immunology Research Center, Avicenna Research Institute, Mashhad University of Medical Sciences (MUMS), Mashhad, Iran
4. Department of Pathology, Omid Hospital, MUMS, Mashhad, Iran

Sincerely yours,

Mohammad R. Abbaszadegan, MT(ASCP), Ph.D.
Associate Professor
Director, Division of Human Genetics
Avicenna Research Institute
Mashhad University of Medical Sciences
Mashhad, Iran postal code: 9196773117
Phone:+98511-7112611-5
Fax: +98511-7112616
abbaszadeganmr@mums.ac.ir